Author's response to reviews

**Title:** Cost-effectiveness analysis of cognitive behaviour therapy for treatment of minor or mild-major depression in elderly patients with type 2 diabetes: study protocol for the economic evaluation alongside the MIND-DIA randomized controlled trial (MIND-DIA CEA)

**Authors:**

Nadja Chernyak (nadja.chernyak@ddz.uni-duesseldorf.de)
Frank Petrak (mail@dr-frank-petrak.de)
Kristin Plack (kplack@uni-mainz.de)
Martin Hautzinger (hautzinger@uni-tuebingen.de)
Matthias J. Müller (mmueller@gmx.de)
Guido Giani (giani@ddz.uni-duesseldorf.de)
Andrea Icks (andrea.icks@ddz.uni-duesseldorf.de)

**Version:** 5 **Date:** 26 June 2009

**Author's response to reviews:** see over
Re: MS 1239143229275878

26. June 2009

Dear Editor,

In accordance with your recommendations we have checked over our manuscript once again in order to avoid any small typographical errors. The following minor changes have been made:

1) Page 6; first paragraph under the subheading “Overview of the MIND-DIA trial”, second line: “diabetes specific” has been changed into “diabetes-specific”.

2) Page 17; second line: “health-care” has been changed into “healthcare “.

3) Page 17; first line under the subheading “Time costs”: “information on” has been removed.

Kind regards, also on behalf of the co-authors,

Nadja Chernyak

Nadja Chernyak MSc
German Diabetes Centre, Leibniz Research Centre at University of Düsseldorf
Institute of Biometrics and Epidemiology
Auf’m Hennekamp 65,
40225 Düsseldorf,
Germany

Tel.: +49 – (0) 211 – 3382 – 404
Fax: +49 – (0) 211 – 3382 – 677
Email: nadja.chernyak@ddz.uni-duesseldorf.de